Author's response to reviews

Title: The relationship between baseline Organizational Readiness to Change Assessment subscale scores and implementation of hepatitis prevention services in substance use disorders treatment clinics.

Authors:

Hildi J Hagedorn (hildi.hagedorn@va.gov)
Paul W Heideman (paulwheideman@gmail.com)

Version: 8 Date: 9 June 2010

Author's response to reviews:

Table 1 has been reformatted and I like it better. However, throughout most of the table the line dividing the subscales begins at the subscale column and then there are a couple of places where the line dividing the subscales begins at the scale column (goes all the way across the page). Since the table has been reformatted I figured it would be easier for you to fix it then for me to recreate the entire table in the new format. Please, let me know if this is ok or if you would like me to recreate the table.

Hildi Hagedorn